Author's response to reviews

Title: Changes in medicine course curricula in Brazil encouraged by the Program for the Promotion of Medical School Curricula (PROMED)

Authors:

Patricia Alves de Souza (p.as@ig.com.br)
Angelica Maria Bicudo Zeferino (ambz@fcm.unicamp.br)
Marco Aurelio Da Ros (ros@ccs.ufsc.br)

Version: 8 Date: 21 November 2008

Author's response to reviews:

The BioMed Central Editorial Production Team

As requested were amended the following points:
• The reference 7 was amended to address an article that contained the same information that are in the text;
• The reference 8 was amended to the Internet to another page that contains the same information contained in the text;
• Amendment of the structure of the abstract;
• Withdrawal of keywords the end of the abstract;
• Amendment of the word introduction to background
• Amendment to the methodology methods;
• Amendments to the Results and Discussion Results and discussion;

Regards,

Patrícia Alves de Souza.